Author's response to reviews

Title: Prognostic value analysis of urokinase-type plasminogen activator receptor in oral squamous cell carcinoma: an immunohistochemical study

Authors:

Roberta Bacchiocchi (bacchiocchi@univpm.it)
Corrado Rubini (c.rubini@univpm.it)
Elisa Pierpaoli (elisa.pierpaoli@gmail.com)
Giulia Borghetti (giuliaborghetti@hotmail.it)
Pasquale Procacci (pasqualeprocacci@hotmail.com)
Pier Francesco Nocini (pierfrancesco.nocini@univr.it)
Andrea Santarelli (andreasantarelli@tele2.it)
Romina Rocchetti (romina.rochetti@libero.it)
Domenico Ciavarella (drdomenicociavarella@yahoo.it)
Lorenzo Lo Muzio (llomuzio@tin.it)
Francesca Fazioli (fazioli@univpm.it)

Version: 3 Date: 28 July 2008

Author's response to reviews:

Dr Robin Cassady-Cain
Assistant Editor
BMC-series Juornals
28 July 2008

MS: 2007313741819333

Dear Dr Cassady-Cain,

Thank you for your mail of July 11, 2008 regarding our manuscript “Prognostic value analysis of urokinase-type plasminogen activator receptor in oral squamous cell carcinoma: an immunohistochemical study” (manuscript 2007313741819333).

As required, we have performed several editorial and grammatical corrections to improve the style of written English.

During revision of the manuscript, I have realized that Dr Giulia Borghetti, who has been involved in the present work for the selection of archived samples and clinical data collection, had been inadvertently omitted in the previous versions. Therefore the authors list has been modified accordingly.

We hope that the manuscript will now be considered acceptable for publication in BMC Cancer.
Sincerely,
Dr Roberta Bacchiocchi